Author's response to reviews

Title: Coordinated care affects hospitalization and prognosis in amyotrophic lateral sclerosis, a chronic neurological disease: a cohort study

Authors:

Valerie Cordesse (valerie.cordesse@psl.aphp.fr)
Florence Sidorok (florence.sidorok@psl.aphp.fr)
Priscilla Schimmel (priscilla.schimmel@psl.aphp.fr)
Josiane Holstein (josiane.holstein@sat.aphp.fr)
Vincent Meininger (vincent.meininger@psl.aphp.fr)

Version: 2 Date: 2 December 2014

Author's response to reviews: see over
Dear Mrs Armée,

Please find attached the modified the article titled “Coordinated care affects hospitalization and prognosis in ALS, a chronic neurological disease: a cohort study”

At your request, the following modifications have been made:

1. Title page provides the modified title of the article, the list the full names, institutional addresses and email addresses for all authors and indicate the corresponding author

2. Line Numbering

In the manuscript have been included line and page numbers. No page breaks have been inserted in your manuscript, including between the Abstract and main article.

3- In the acknowledgment section have been included:

a- a section on competing interests both financial and non-financial

b- a section on Authors’ contributions describing the contributions of each author

4. Figures: The figure included in the previous manuscript has been deleted

Yours sincerely,

Vincent Meininger, MD, PhD.

Professor of Neurology Université Pierre et Marie Curie

Medical Practitioner, Hôpital de la Salpêtrière

47 Boulevard de l'Hôpital, 75013 Paris,

France

E-mail: vincent.meininger@psl.aphp.fr

Tel: +33-680451730

Fax: +33-44243269